

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Alós-Ferrer, Carlos; Buckenmaier, Johannes; Kirchsteiger, Georg

Working Paper Do traders learn to select efficient market institutions?

Working Paper, No. 364

Provided in Cooperation with: Department of Economics, University of Zurich

Suggested Citation: Alós-Ferrer, Carlos; Buckenmaier, Johannes; Kirchsteiger, Georg (2020) : Do traders learn to select efficient market institutions?, Working Paper, No. 364, University of Zurich, Department of Economics, Zurich, https://doi.org/10.5167/uzh-190321

This Version is available at: https://hdl.handle.net/10419/225013

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



University of Zurich

Department of Economics

Working Paper Series

ISSN 1664-7041 (print) ISSN 1664-705X (online)

Working Paper No. 364

Do Traders Learn to Select Efficient Market Institutions?

Carlos Alós-Ferrer, Johannes Buckenmaier and Georg Kirchsteiger

September 2020

Do Traders Learn to Select Efficient Market Institutions?*

Carlos Alós-Ferrer[†] University of Zurich Johannes Buckenmaier[‡] University of Zurich

Georg Kirchsteiger[§] Université Libre de Bruxelles

This Version: September 2020

Abstract

When alternative market institutions are available, traders have to decide both where and how much to trade. We conducted an experiment where traders could decide to trade either in an (efficient) double-auction institution or in a postedoffers one, which should favor sellers. When sellers face decreasing returns to scale (increasing production costs), fast coordination on the double-auction occurs, with the posted-offers institution becoming inactive. In contrast, under constant returns to scale, both institutions remain active and coordination is slower. The reason is that, in a finite-horizon setting, sellers trade off larger efficiency in a market with dwindling profits for biased-up profits in a market with vanishing customers. Hence, our results indicate that efficiency alone might not be sufficient to guarantee coordination on a single market institution if the distribution of the gains from trade is asymmetric. Trading behavior approaches equilibrium predictions (market clearing) within each institution, but switching behavior across institutions is explained by simple rules of thumb, with buyers chasing low prices and sellers considering both prices and trader ratios.

JEL Classification: $D4 \cdot D83 \cdot L1$

Keywords: Market Selection \cdot Market clearing \cdot Posted offer market \cdot Constant returns to scale \cdot Experiment

Working Paper. This is an author-generated version of a research manuscript which is circulated exclusively for the purpose of facilitating scientific discussion. All rights reserved. The final version of the article might differ from this one.

 $^{^*{\}rm Financial}$ support from the German Research Foundation (DFG) through project AL-1169/5-1 is gratefully acknowledged.

[†]Corresponding author: carlos.alos-ferrer@econ.uzh.ch. Department of Economics, University of Zurich. Blümlisalpstrasse 10, 8006 Zurich, Switzerland.

[‡]Department of Economics, University of Zurich. Blümlisalpstrasse 10, 8006 Zurich, Switzerland.

ECARES, Université Libre de Bruxelles, Avenue F. D. Roosevelt 50, CP114, 1050 Brussels, Belgium

1 Introduction

When investigating the functioning of markets one is puzzled by the huge variety of institutional rules that govern them. Financial assets are traded e.g. on call markets as well as with continuous double auctions (see Madhavan, 1992). Real estate is sold at auctions (Quan, 1994), through brokers (Zumpano et al., 1996), and by means of direct negotiations. Consumers increasingly buy all sort of goods in centralized online platforms, but retail stores survive. More and more, trader decisions start not with the question of how much to buy or sell, but with the question of where to trade.

If all market institutions lead to market clearing, enough traders are always present, and there are no large differences between the traders flocking to one or the other institution, the multiplicity of the latter should matter little. On the contrary, a large body of theoretical, empirical, and experimental evidence strongly suggests that institutional differences among market institutions are far from being inconsequential (see, among many others, Lucking-Reiley, 2000; Reynolds and Wooders, 2009; Ariely et al., 2005; Ausubel et al., 2014; Genesove and Hansen, 2016). In particular, the properties of market institutions determine whether and how fast actual prices and quantities reach market clearing level (see, e.g., Plott, 1982). And yet, when different market institutions are legally feasible, as it is the case for most goods, it is far from clear how actually-used market institutions come to be, and what characterizes the ones that traders favor. It has been argued that the actual market institutions are the most efficient ones given the characteristics of the good. Hayek (1967) defended the proposition that traders, if left to their own devices, choose the most efficient trading institution that the good at hand allows for.¹ This paper aims to test this conjecture experimentally. Specifically, we set out to investigate whether efficient institutions are indeed used in the long run, while inefficient ones fade as they are eventually avoided by the traders.

Take a situation where different trading platforms are feasible for trading a particular good. Each trader has to choose which one he/she wants to use.² An individual buyer or seller will not choose a platform where he/she cannot find a trading partner, even if the institutional setup of the avoided platform would lead to very efficient trading outcomes. Hence, traders face a (partial) coordination game when choosing on which platform they want to trade.³ After choosing the trading institution, i.e. for a given distribution of buyers and sellers over the feasible platforms, the institutional setups—together with demand and supply resulting from the platform choice—will lead to more or less efficient trading outcomes. Hence, the whole process is characterized by two steps: First, traders have to choose the trading platform (where to trade). Then, they conduct

¹This "Hayekian" position can also be discerned in the writings of Klein and Alchian (1978) and Williamson (1975).

 $^{^{2}}$ In our notation sellers are female and buyers are male.

³Coordination games have been widely studied both theoretically and experimentally (see, e.g. Young, 1993; Van Huyck et al., 1997). To the best of our knowledge, though, these investigations have not been connected to the functioning of markets.

their trades with the partners available at their platform (how much and at which price to trade).

In our experiment each subject is either a buyer or a seller. In the first stage of each experimental round, each subject had to choose individually between a posted offer (PO) institution and an open double auction (DA, Plott and Gray, 1990; "open" meaning that there was no auctioneer, thus multiple prices could be realized). In the second stage of each round, subjects could trade multiple units of a homogeneous good according to the rules of the chosen institution with those potential trading partners who had opted for the same institution. At the end of each round, all the participants received aggregate information about what happened at both institutions, enabling them to make an informed decision on where to trade in the next round: number of buyers and sellers, distribution of trading prices (and their average), and number of trades per buyer and per seller at each institution. Then a new round started, where first all subjects had to choose between the DA and the PO. After that, they traded on the chosen platform anew, received information about the outcome of both institutions, and so on.

For each trade conducted, the seller received the price paid by the buyer, but in turn she had to pay production costs that were determined by the experimental design. Hence, her net earnings from a trade were the difference between the price and production costs. For a buyer, the net earnings from a trade were the difference between the resale values, which were determined by the design of the experiment, and the price paid. That is, in each given round, the actual, endogenously-induced supply and demand at a given platform was determined by the production costs and resale values of the sellers and buyers who opted for that platform.

We chose DA and PO as institutional alternatives since it is very well documented that, when functioning in isolation, DAs are highly efficient. Theoretically, for example, Rustichini et al. (1994) show that, in double auctions, possible inefficiencies due to traders' strategic misreporting vanish quickly as the number of traders increase. Empirically, it has been repeatedly shown that DAs induce a very quick convergence of prices and quantities to market clearing levels (see, e.g., the seminal paper of Plott and Smith, 1978a). As a consequence, the DA has been found to be the most efficient market institution when efficiency is measured by the sum of all gains from trade reaped by buyers and sellers (as it is usually done in the context of market experiments). In fact, DA is routinely used in market experiments as a proxy for competitive markets (e.g., Plott, 2000; Crockett et al., 2011; Gillen et al., 2020). On the other hand, in the PO prices and quantities typically show a much slower convergence to the market clearing level, with prices typically converging from above (see Plott and Smith, 1978a). For given demand and supply, the efficiency level is considerably higher for a DA than for a PO. If efficiency drives the selection of market institutions, we should expect that traders coordinate on the DA, and that the PO eventually falls into disuse.

The experimental results indicate that the validity of this prediction depends crucially on the properties of the distribution of production costs. If the production technology

underlying the sellers' supply function exhibits decreasing returns to scale, traders indeed learn to coordinate on the efficient DA, with PO becoming mostly inactive in finitely many rounds. In contrast, if production displays constant returns to scale, resulting in a flat induced supply function, coordination on DA happens more slowly and both institutions remain active until the last round.⁴ This difference might be caused by the fact that a Pareto-dominance relation between the two institutions depends on the distribution of the production costs. For a production technology with constant returns to scale, sellers' earnings are (nearly) zero at an institution like the DA that induces prices which are at (or very close to) the market clearing level. Hence, one market side, i.e. the sellers, has a strong incentive to try to coordinate traders on an institution like the PO, where prices are above the market clearing level. To put it differently, if an institution yields outcomes which Pareto-dominate those of competing ones, it should be expected that it will clearly and quickly drive the latter out of the market. However, switching from coordination on the less efficient PO to coordination on the more efficient DA does not constitute a Pareto-improvement if sellers produce under constant returns to scale. Our results thus suggest that efficiency alone, as measured by the sum of the gains from trade, might not be enough to guarantee that a market institution drives competing ones out of the field. The reason for this is that sellers face a tradeoff between a more efficient institution which brings their profits down and an inefficient, low-volume one which however typically yields higher prices. In the absence of Pareto-dominance, the distribution of gains from trade, and not only aggregate efficiency, becomes consequential for market selection and survival.

Our data allows us to look both at actual trading behavior within an institution and the previous decision of which institution to trade in. Regarding the former, we observe that aggregate behavior, as reflected by actual prices, does converge to the theoretical market-clearing benchmark both in PO and in DA (even though we use an open DA implementation without an auctioneer). Regarding the choice of market institution, we find that simple rules of thumb are sufficient to capture most behavior. Buyers seem to mostly chase after low (past) prices, switching to the institution where observed prices were better for them, along the lines of previously-postulated behavioral rules based on past performance only (e.g. Huck et al., 1999; Offerman et al., 2002; Bosch-Domènech and Vriend, 2003). Sellers' behavior appears to be more complex, but is well-explained by the combination of a similar rule which points toward high prices, and a complementary rule which focuses on favorable trader ratios. In our setting, where all buyers have the same resale values and all sellers have the same cost functions, equilibrium prices are a function of trader ratios, hence the latter rule might reflect forward-looking behavior.

An extensive experimental literature, going back to Chamberlin (1948) and Smith (1962), has analyzed the empirical properties of market institutions for given demand and

⁴This result is similar in flavor to those of Alós-Ferrer and Kirchsteiger (2015, 2017). In these theoretical papers we found that traders who have to learn which platform to use are more likely to coordinate on a market clearing institution if sellers have an increasing supply function. See also Alós-Ferrer et al. (2010) for implications for market design.

supply when viewed in isolation (for an overview see Plott and Smith, 1978b, Part 1).⁵ Surprisingly, however, very few contributions have combined the experimental investigation of actual trading behavior with the analysis of the *choice* of the trading institution for a given good. One notable exception is Campbell et al. (1991), which investigated the endogenous choice between a computerized double-auction market (with an auctioneer) and (illegal) off-floor trading in the context of stock markets, and its impact on the bid-ask spread. The latter was implemented as direct negotiations, specifically traders could submit direct offers for blocks of three units to their two neighboring traders of the other market side. Kugler et al. (2006) compare direct negotiations and centralized markets in an experimental setting where each trader can trade a single unit only. Trade can be conducted in two alternative institutions. The first captures centralized markets through a sealed-bid double auction with a single market clearing price (as in Campbell et al., 1991, computed by an auctioneer). The second captures direct negotiations through bilateral matching, where the matching was designed as to maximize trade. Inefficiency might appear due to asymmetric distribution of traders across the two institutions, but their focus was not on the comparison of institutions in terms of efficiency. Rather, they focus on heterogeneous traders with different values and find that different types of traders generally prefer different market mechanisms. Their results show an unraveling of direct negotiations, which is led by higher-value traders (buyers with high resale values or sellers with low costs), who learn faster to coordinate on the centralized market. Last, the theoretical work of Alós-Ferrer and Kirchsteiger (2015) included an experiment on platform selection but adopted a reduced-form payoff table for actual trade, that is, traders did not make actual trading decisions. Alós-Ferrer and Kirchsteiger (2015) focused on the selection of market-clearing institutions (vs. institutions with price biases) and showed that certain alternative institutions could survive in the long run when players followed simple behavioral rules of thumb.

The rest of the paper is organized as follows. Section 2 describes the experimental setting and the data. Section 3 briefly reviews the formal framework. Section 4 analyzes convergence and coordination on our market institutions. Section 5 examines how endogenous market selection affects efficiency within each institution and which market side is able to reap the gains from trade. Section 6 shows that simple behavioral rules can capture switching behavior across institutions and the differences between buyers and sellers. Section 7 concludes.

2 The Experiment

A total of N = 480 subjects (258 females, mean age 25.3 years, SD = 6.9) participated in 15 experimental sessions. Subjects were recruited from the student population of the University of Cologne (excluding students majoring in psychology) via ORSEE (Greiner,

⁵Williams et al. (2000) examined experimental double-auction markets in a setting where participants traded two different goods concurrently, with a different market for each good.

2015). All interactions took place via a custom-made market interface (see Section 2.2) programmed in zTree (Fischbacher, 2007). There were two treatments, DRS and CRS (see Section 2.3), with 15 groups (240 subjects) randomly assigned to each of them.

2.1 Procedures

A session comprised 32 subjects randomly allocated into two groups of 16 subjects each. Subjects interacted only with other subjects within their own group. That is, each group constitutes an independent observation. Within each group half of the subjects were randomly assigned the role of a buyer and half were assigned the role of a seller. Roles remained fixed throughout the experiment.

Before the beginning of the experiment, subjects received written instructions describing the course of the experiment and the market selection stage (see Appendix). These general instructions were also read aloud by the experimenter. After that, subjects received specific written instructions according to their assigned role (buyer or seller) with a detailed description of how trade was conducted at each of the two platforms. Subjects then answered four control questions to ensure their understanding of the experimental environment.

The instructions also contained a detailed description of the payment procedure, which was carried out truthfully. At the end of the experiment a subject's earnings from each round were added up and converted to euros at a rate of $\in 1.5$ for 100 experimental currency units. In addition subjects received a show up fee of $\in 4$ leading to an average total remuneration of 23.68 EUR. Sessions lasted about 105 minutes on average.

2.2 Design and market interface

All market interactions took place via a custom-made interface programmed in zTree (Fischbacher, 2007). There were 25 trading rounds, each consisting of two sequential stages. In the first stage (Market Selection), each trader chose individually between two market platforms, a double auction (DA) and a posted offer (PO) institution. In the second stage (Trading), subjects could trade at the selected platform according to the rules of that institution, interacting with all traders who selected the same market. That is, the number of traders within each market was determined endogenously and, hence, so was demand and supply. In any given round traders had to commit to the selected market and could only trade within that market, but they could freely switch markets between rounds.

Each population of traders consisted of 8 buyers and 8 sellers. Roles were assigned randomly at the beginning of the experiment and stayed fixed throughout. We use a multi-unit setup, specifically, each trader can trade up to 6 units of a homogeneous good. Buyers received exogenously-given resale values for each unit bought. Sellers faced exogenously-given production costs for each unit sold.⁶ Buyers and sellers were homogeneous, that is, resale values and production costs were the same for all buyers and sellers, respectively. For each trade conducted, the seller received the price at which the trade took place and had to pay the corresponding production costs, whereas buyers had to pay the price and received the corresponding resale value. Endowments and capacities were reset each period, that is, the experiment involved stationary repetition as standard in market experiments.

In each session, the trading rounds were preceded by two trial rounds intended to give subjects the opportunity to familiarize themselves with the interface and the rules of trade at both platforms. The trial periods featured no market selection stage. Instead, in the first trial round 4 buyers and 4 sellers were randomly assigned to each platform, and in the second trial round platform allocation was reversed.

Market Selection. In the first stage, traders chose between a DA and a PO institution. In all but the first round, subjects received aggregate information regarding the performance of both platforms (irrespective of which platform they had selected previously) as well as a summary of their own performance in the previous round. Specifically, aggregate information was presented in a table showing the number of buyers and sellers, the average trading price, and the number of trades per buyer and seller separately for each institution.⁷ Additionally, subjects were provided with a price-quantity histogram for each market indicating the number of units traded at each price. Own performance was summarized via a second table listing the prices of all units traded by the subject and the overall profit in that trading round. On the same screen subjects then could choose one of the two platforms, neutrally labeled "Market A" and "Market B," via a button press. Subjects had 30 seconds for this choice and the remaining time was shown on screen. If a subject failed to select a platform within the time limit, he/she could not trade in the next trading round.⁸

Double auction institution. Figure 1 displays a graphical illustration of the interface for DA (for buyers). In DA all traders could simultaneously make offers and accept open offers by traders from the other market side. That is, there was no auctioneer or market-maker, and all offers were visible for all traders. Traders could offer any number of units $q \in \{1, \ldots, q_r\}$, with $q_r \leq 6$ being the remaining number of units that could be traded, at any price p by submitting an offer of the form (p, q), subject to the constraint of not incurring a net loss within a trading period.⁹ At any point in time a trader could have

 $^{^{6}}$ In this sense, preferences in our experiment were induced. Crockett et al. (2020) make the point that, in market experiments where traded goods involve exogenous risk, it might be important to differentiate between induced and elicited preferences. However, the trading stage of our experiment involved no exogenous risk and followed the tradition of Chamberlin (1948), Smith (1962), and many others.

⁷See Figure C.1 in the Appendix for a screenshot.

⁸This happened for 22 of the 480×25 market-selection decisions, i.e. around 0.2% of the time.

⁹The range of feasible offers that could be submitted and the feasible active offers that could be accepted by each trader were limited in such a way that net within-period losses were not possible. That



Figure 1: Schematic representation of the screen layout for DA (buyers).

at most one active offer. However, offers could be withdrawn or replaced by new offers at any time. A submitted offer remained active until it was withdrawn or replaced by a new offer. Further, traders could always accept active offers by the other market side. All traders could see all open offers from both buyers and sellers (see Figure 1). Offers from the own and the other market side were always shown on the left and right parts of the screen, respectively. Each offer (p,q) was represented by q boxes showing the price p with each box representing an offer to buy/sell one unit at that price (see Figure 1). Traders could accept offers from the other market side, that is buy/sell one unit at the shown price, by clicking on the corresponding box. Traded units were indicated by crossed-out boxes and remained visible until the end of the trading round. Each subject could trade up to a maximum of 6 units and the remaining number of units that could be traded at any point was shown on screen.¹⁰ For Additionally, buyers, respectively sellers, were shown the resale values, respectively production costs, for each of the six units as well as the price of already-traded units. Subjects could leave the trading stage by pressing the "leave market stage" button, in which case any still-active offer was withdrawn. The maximum duration of a trading round at DA was 90 seconds, with the remaining time shown on screen. Alternatively, the market platform was closed if no further trade was possible (for example because all buyers had left the trading stage or all sellers were out of stock).

Posted offer institution. In PO buyers and sellers moved sequentially in two distinct stages.¹¹ In the fist stage, sellers simultaneously submitted offers of the form (p, q)

is, single units could be traded at a loss as long as this was compensated by profits from other trades in the same trading period. Subjects received a detailed explanation of this procedure in the written instructions.

¹⁰See Figure C.2 in the Appendix for a screenshot.

 $^{^{11}\}mathrm{See}$ Figure C.3 in the Appendix for screen shots.

	DRS		CRS	
Unit	Resale Values	Costs	Resale Values	Costs
1	80	15	80	40
2	65	35	65	40
3	60	40	60	40
4	55	50	55	40
5	40	60	40	40
6	35	65	35	40

Table 1: Baseline resale values and production costs for treatments DRS and CRS.

proposing to sell q units at price p.¹² Sellers were shown their production costs for all six units, but received no information about the offers made by other sellers during this stage. This first stage lasted for a maximum of 20 seconds, or, alternatively until all sellers had committed to an offer. At the end of this stage the offers made by all sellers were collected and the second stage began. In that stage, buyers moved sequentially in a randomly-determined order. When a buyer's turn came, he observed all offers made in the first stage, as well as all units bought by previous buyers, and could buy units by clicking on still-available offers. As in the DA case, available offers were indicated by numbered boxes and already-traded units by crossed-out boxes (as in the right part of Figure 1).¹³ A buyer's turn ended after 20 seconds or once no further units were available. The trading round ended once all buyers had had their turn, or no further units were available.

2.3 Treatments

In a given round, the resale values and production costs of the traders that opted for a particular institution determined the induced supply and demand at that institution. There were two treatments, DRS and CRS, which correspond to decreasing and constant returns to scale for the supply side of the market, respectively. Specifically, the treatments differed in the resale values and production costs as given in Table 1.¹⁴ In both treatments the demand function induced via buyers' resale values was decreasing in the price. This was accomplished by setting decreasing resale values for each unit traded by a buyer. In treatment DRS, sellers faced increasing production costs (decreasing returns to scale), hence the induced supply function was increasing in the price. In treatment CRS, sellers faced a constant production cost instead (constant returns to scale), and as a consequence the induced supply function was flat. In each round a common shock

¹²As in the DA, feasible offers were limited to (p,q) pairs such that, if all offered units were traded, no net within-period loss for the seller would result.

¹³Similarly to the DA case, buyers could only accept offers not leading to a net within-period loss.

¹⁴The trial periods used different resale values and production costs.

 X_t was added to all these baseline resale values and production costs, so that the exact numbers varied from round to round, but induced demand and supply were unaffected.

3 Equilibrium Prices and Market-Clearing Benchmark

In this section, we consider the theoretical benchmark where markets clear, taking as given an allocation of traders to institutions, and derive equilibrium market prices and total trader surplus, which we will use as a benchmark to measure efficiency.

Fix an institution, and consider the population of traders that has chosen to trade at that institution. This population consists of n buyers and m sellers (with $n, m \in \{0, \ldots, 8\}$ in the experiment). Each trader can trade up to Q units (Q = 6 in the experiment) of a single homogeneous good. Denote the price of the good by $p \ge 0$. A typical buyer is characterized by a vector of resale values (r_1, \ldots, r_Q) with $r_1 \ge$ $\cdots \ge r_Q > 0$ where r_k is the buyer's resale value for the kth unit. A typical seller is characterized by a vector of production costs (c_1, \ldots, c_Q) with $0 < c_1 \le \cdots \le c_Q$ where c_k is the seller's production cost for the kth unit. Assume that $r_1 \ge c_1$, that is, beneficial trade is possible for any $p \in [r_1, c_1]$. For a given price p, the resale values induce a (weakly) decreasing demand function given by

$$d(n,p) = nk$$
 where $k = \max\{0, \sup\{k \in \{1, \dots, Q\} \mid p \le r_k\}\}.$

That is, demand is given by the largest number of units that the *n* buyers are willing to purchase (implicitly assuming, for simplicity, that traders are willing to trade when exactly indifferent). Specifically, if $r_Q ,$ *k*is the unique number such that $<math>r_{k+1} . If <math>p \leq r_Q$, then k = Q and demand is given by nQ (the capacity constraint becomes binding). If $p > r_1$, the set over which the supremum is taken above is empty (so the supremum is formally $-\infty$), and hence k = 0 and demand is nonexistent. The induced demand function is illustrated in Figure 2 for both treatments in the experiment, for the case where all 8 buyers are at the same institution.

Analogously, for a given price p, the production costs induce a (weakly) increasing supply function given by

$$s(m, p) = mk$$
 where $k = \max\{0, \sup\{k \in \{1, \dots, Q\} \mid p \ge c_k\}\}$.

That is, supply is given by the largest number of units that the *m* sellers are willing to sell. Specifically, if $c_1 ,$ *k* $is the unique number such that <math>c_k \leq p < c_{k+1}$. If $p \geq c_Q$, then k = Q and supply is given by mQ (the capacity constraint becomes binding). If $p < c_1$, k = 0 and supply is nonexistent. The induced supply function is illustrated in Figure 2 for both treatments in the experiment, for the case where all 8 sellers are at the same institution.

In our discrete setting, demand and supply are step functions and hence equilibrium prices might correspond to an interval instead of a unique value. Let $p(m, n) = \sup\{p \mid$



Figure 2: Induced demand and supply functions for both treatments (DRS left, CRS right) in case of full coordination on the respective institution.

 $d(n,p) \geq s(m,p)$ and $\bar{p}(m,n) = \inf\{p \mid d(n,p) \leq s(m,p)\}$. It follows that $\underline{p}(m,n) \in \{-\infty, c_1, \ldots, c_Q\}$ and $\bar{p}(m,n) \in \{+\infty, r_1, \ldots, r_Q\}$. Further, note that $\underline{p}(m,n) \leq \bar{p}(m,n)$, and, since $r_1 \geq c_1$, at least one of $\bar{p}(m,n)$ and $\underline{p}(m,n)$ must be finite.

The set of equilibrium prices $p^*(m, n)$ is given by

$$p^*(m,n) = \begin{cases} [\underline{p}(m,n), \overline{p}(m,n)] & \text{if } \underline{p}(m,n), \ \overline{p}(m,n) \text{ are finite} \\ \{r_1\} & \text{if } \overline{p}(m,n) = \infty \text{ (that is, } n \ge mQ) \\ \{c_1\} & \text{if } \underline{p}(m,n) = -\infty \text{ (that is, } m \ge nQ). \end{cases}$$

Figure 2 illustrates $p^*(m,n)$ for both treatments in the experiment, for the particular case m = n = 8. Note that $\bar{p}(m,n) = \infty$ implies that $n \ge mQ$ and demand always exceeds supply, hence r_1 is the unique equilibrium price at which trade actually occurs. Analogously, if $\underline{p}(m,n) = -\infty$, $m \ge nQ$, supply always exceeds demand, and c_1 is the unique equilibrium price at which trade occurs.

In our experiment, the resale values were given, and the production costs varied across treatments (DRS and CRS). Tables B.1 and B.2 in the Appendix report the equilibrium price intervals for both treatments. In the analysis below, and as a first measure of the efficiency of the institution, we will report the distance from actually-realized trading prices per round to the benchmark equilibrium price intervals $p^*(m, n)$.

As a second measure of efficiency, we consider the largest achievable total trader surplus for a given number of buyers and sellers. Let the mQ resale values of the m



Figure 3: Top: Average (over 15 markets) number of buyers/sellers at DA per period for DRS (left) and CRS (right). Bottom: Average (over 15 markets) fraction of markets where DA/PO is active per period for DRS (left) and CRS (right).

buyers be $R_1 \ge \cdots \ge R_{mQ}$ and let the nQ costs of the sellers be $C_1 \le \cdots \le C_{nQ}$. The maximum gains of trade are defined as the largest-possible total trader surplus, that is

$$S^*(m,n) = \sum_{i \in I} (R_i - C_i) \text{ where } I = \{i \mid R_i > C_i, i \le \min\{nQ, mQ\}\}.$$

Figure 2 illustrates $S^*(m, n)$ for both treatments in the experiment, for the particular case m = n = 8. The values of $S^*(m, n)$ for DRS and CRS are given in Tables B.3 and B.4 in the Appendix, respectively. In the analysis below, we will compare the actuallyrealized total trader surplus (defined as the sum of the differences between resale value and production cost for all actually-traded units) to $S^*(m, n)$.

4 Results: Market Selection

4.1 Decreasing returns

We first consider treatment DRS, where the supply side faced decreasing returns to scale (N = 240 in 15 market observations). The top-left panel of Figure 3 shows the average number of buyers and sellers at DA for each period. We see a clear convergence

	Number of	buyers at DA	Number of	of sellers at DA	
	1	2	3	4	
Period	0.0734^{***}	0.0636^{***}	0.0959^{***}	0.0864^{***}	
	(0.0074)	(0.0074)	(0.0088)	(0.0090)	
RatioDA		-0.5988^{***}		0.1776	
		(0.1308)		(0.1373)	
Constant	5.8000^{***}	6.5959^{***}	5.3580^{***}	5.3413^{***}	
	(0.3470)	(0.2954)	(0.3790)	(0.3387)	
R^2 (overall)	0.1193	0.1652	0.1663	0.1194	
Observations	375	359	375	359	
Markets	15	15	15	15	

Table 2: Linear random effects regressions for DRS.

Notes: Robust standard errors in parentheses. * p < 0.1, ** p < 0.05, *** p < 0.01.

toward DA over time.¹⁵ That is, as time goes by, more and more traders of each market side choose the DA. This is confirmed in a series of linear random effects regressions on the number of buyers/sellers at DA per period, as reported in Table 2.¹⁶ The variable RatioDA is the (lagged) buyer-seller ratio at DA. The regressions show that buyers are attracted by a low buyer-sellers ratio, but the latter has no effect on sellers' behavior.

The bottom-left panel of Figure 3 shows the fraction of markets where DA and PO are active over time. Across all 15 markets the DA is active in all but one period for a single market, that is, on average the DA is active 99.7% of the time. In contrast, the PO is only active 48% of the time. The average fraction of periods in which DA is active is significantly larger than the average fraction of periods in which PO is active (Wilcoxon signed-rank test, WSR: N = 15, z = 3.352, p < 0.001). We obtain a similar result for the average number of buyers or sellers, that is, on average there are more buyers and more sellers at DA than at PO (WSR, buyers: N = 15, z = 3.352, p < 0.001; sellers: N = 15, z = 3.352, p < 0.001).

To confirm that traders increasingly coordinate on DA over time, we compare coordination at the beginning against coordination at the end of the experiment. To that end, we exclude the first period and split the remaining 24 periods in three parts of 8 periods each. We first consider the fraction of time DA or PO are active in part 1 vs. part 3. Since DA is essentially active in all periods, there is no difference for DA between parts 1 and 3 (WSR, N = 15, z = -1.000, p = 0.317). On the other hand, PO is active more often in part 1 (64.2% of the time) than in part 3 (28.3%; WSR, N = 15, z = 3.048, p = 0.002). A similar picture emerges when looking at the number of buyers and sellers at DA, which is significantly larger at the end of the experiment compared to the beginning (WSR, buyers: N = 15, z = -3.311, p < 0.001; sellers: N = 15, z = -3.411, p < 0.001).

¹⁵Figure A.1 in the Appendix displays the analogous graphs for each market session.

¹⁶A fractional logit regression yields the same results. See Table A.1 in the Appendix.

	Number of	buyers at DA	Number of	sellers at DA
	1	2	3	4
Period	0.0882^{***}	0.0836^{***}	0.0622^{***}	0.0548^{***}
	(0.0078)	(0.0078)	(0.0101)	(0.0107)
RatioDA		-0.5331^{***}		0.3603^{**}
		(0.1127)		(0.1558)
Constant	5.1153^{***}	5.7994^{***}	5.0507^{***}	4.7695^{***}
	(0.3315)	(0.2999)	(0.3159)	(0.3279)
R^2 (overall)	0.1629	0.1870	0.0715	0.0495
Observations	375	360	375	360
Markets	15	15	15	15

Table 3: Linear random effects regressions for CRS.

Notes: Robust standard errors in parentheses. * p < 0.1, ** p < 0.05, *** p < 0.01.

4.2 Constant returns

We now consider treatment CRS (N = 240 in 15 different market observations). The top-right panel of Figure 3 illustrates the evolution of the number of buyers and sellers at DA over the course of the experiment in CRS. We again observe convergence toward DA, although it appears to be slower than in the case of DRS (see next subsection for a treatment comparison).¹⁷ Table 3 displays a series of linear random effects regressions of the number of buyers/sellers at DA on period and shows that traders move toward DA over time.¹⁸ Buyers are attracted by a low buyer-seller ratio, and the opposite is true for sellers.

The bottom-right panel of Figure 3 shows the fraction of markets in CRS where DA and PO are active over time. DA is always active, whereas PO is only active 66.4% of the time, which is significantly less often than DA (WSR: N = 15, z = 3.333, p < 0.001). Buyers and sellers favor DA over PO, that is, on average there are both more buyers and more sellers at DA (WSR, buyers: N = 15, z = 3.351, p < 0.001; sellers: N = 15, z = 3.352, p < 0.001).

As in the previous treatment, we compare coordination in parts 1 (periods 2–9) and 3 (periods 18–25). We first consider the fraction of time DA or PO are active in part 1 vs part 3 of CRS. Since in CRS DA is always active in all periods, there is no difference for this platform. On the other hand, PO is active more often in part 1 (78.3% of the time) than in part 3 (59.2%; WSR, N = 15, z = 2.094, p = 0.036). The analogous statement holds for the number of buyers and sellers at DA, which is significantly larger in part 1 compared to part 3 (WSR, buyers: N = 15, z = -2.902, p = 0.004; sellers: N = 15, z = 2.907, p = 0.004).

¹⁷Figure A.2 in the Appendix displays the analogous graphs for each market session.

 $^{^{18}\}mathrm{A}$ fractional logit regression yields the same results. See Table A.2 in the Appendix.

	Number of	buyers at DA	Number of sellers at DA		
	Rounds 2-9	Rounds 18-25	Rounds 2-9	Rounds 18-25	
Period	0.1536^{***}	0.0365^{**}	0.1849^{***}	0.0278	
	(0.0351)	(0.0178)	(0.0412)	(0.0268)	
CRS	-0.6583	-0.4500	-0.5167	-1.1000^{**}	
	(0.4058)	(0.4318)	(0.4146)	(0.4613)	
Constant	5.3554^{***}	6.5484^{***}	4.8746^{***}	6.6528^{***}	
	(0.3550)	(0.4677)	(0.3593)	(0.6150)	
R^2 (overall)	0.0864	0.0346	0.0752	0.1238	
Observations	240	240	240	240	
Markets	30	30	30	30	

Table 4: Regressions for comparison DRS vs CRS in first and last part.

Notes: Robust standard errors in parentheses. * p < 0.1, ** p < 0.05, *** p < 0.01.

4.3 Treatment comparison

We now compare the results of treatment CRS with those of the DRS treatment. Examination of Figure 3 suggests that convergence toward coordination on DA is slower in the case of constant returns to scale. A treatment comparison shows that, indeed, there are on average more buyers and more sellers at DA in DRS than in CRS (Mann-Whitney-U test, MWU, buyers: N = 30, z = 1.660, p = 0.097; sellers: N = 30, z = 2.158, p = 0.031).¹⁹

The differences are also reflected in the comparison between the beginning and the end of the experiment. While there is no difference in activity for DA or PO in part 1 (MWU, DA: N = 30, z = -1.000, p = 0.317; PO: N = 30, z = -1.272, p = 0.204), in part 3 PO is marginally more active in CRS than in DRS (MWU, N = 30, z = -1.721, p = 0.085). Activity is, however, a somewhat coarse measure, hence we turn to the number of buyers and sellers at DA as a more fine-grained measure of convergence. With this measure, we find that already at the beginning of the experiment more buyers and sellers coordinate on DA in DRS compared to CRS (MWU; buyers, N = 30, z = 1.808, p = 0.071; sellers, N = 30, z = 1.787, p = 0.074). The difference subsists at the end of the experiment (buyers, N = 30, z = 0.969, p = 0.097; sellers, N = 30, z = 2.926, p = 0.003).

Table 4 displays the results of linear regressions comparing the number of buyers and sellers in the last part of the experiment (round 18-25) across treatments. That is, the dummy CRS takes the value 1 for market observations in that treatment. We observe that the number of sellers at DA is significantly smaller for CRS market observations, but only in part $3.^{20}$

¹⁹PO is also active more often in CRS (66.4% of the time) than in DRS (48.0%), but the difference fails to reach significance (MWU, N = 30, z = -1.519, p = 0.129).

²⁰A fractional logit regression yields the same results. See Table A.3 in the Appendix.



Figure 4: Average (over 15 markets) of the differences between average realized prices and the equilibrium price $(p - p^*)$, or the difference with the closest value in the interval $p^*(m, n)$ at PO vs. DA per period

5 Results: Efficiency and the Gains from Trade

The previous section has concentrated on the market selection decision and shown that traders learn to gradually coordinate on the DA platform, although convergence is stronger under decreasing returns to scale. This section examines trading decisions within each institution, and asks whether actual trade is efficient and which market side is able to reap larger shares of the gains from trade.

5.1 Trading Efficiency and Market-Clearing

We consider two dimensions of efficiency of an institution. The first is the difference between the actual price and the equilibrium price interval $p^*(m, n)$ given that n buyers and m sellers are currently present at that institution. The second is the fraction of realized gains (trader surplus) relative to the maximal possible gains of trade $S^*(m, n)$.

Figure 4 shows the evolution of the average price at both institutions relative to the equilibrium price, for both treatments. For illustrative purposes, what is plotted is the difference $p - p^*$, where p is the realized average price and p^* is the equilibrium price, if the latter is unique. If $p^*(m,n)$ is a proper interval, we plot the difference with the closest value in that interval.

To evaluate how close realized prices were to equilibrium ones, we consider the distance, i.e. the (average across markets of the) absolute value of the average difference each period. In DRS, realized prices where closer to equilibrium in DA (average distance 2.86) than in PO (9.45; WSR, N = 15, z = -3.294, p < 0.001). This is also true in CRS, although the difference is smaller. In this treatment, the average distance between realized and equilibrium prices was 7.20 for DA and 9.25 for PO (WSR, N = 15, z = -2.499, p = 0.013). Comparing treatments, for DA the distance to equilibrium prices is larger



Figure 5: Average (over 15 markets) realized gains (as fraction of maximal possible gains) at PO vs. DA per period.

in CRS than in DRS (MWU, N = 30, z = -4.500, p < 0.001), whereas we find no significant difference across treatments for PO (N = 30, z = -0.477, p = 0.633).

Figure 5 displays the evolution of the average fraction of the maximal gains of trade which were actually realized in both treatments and for both institutions. In DRS the double auction realizes on average 95.8% of the maximum possible gains of trade, whereas in PO realized gains are only 71.2%. The difference is highly significant (WSR, N = 30, z = 3.408, p < 0.001). In CRS, DA realizes 98.1% of all possible gains, while in PO realized gains are only 83.5%. Again, the difference is highly significant (WSR, N = 30, z = 3.408, p < 0.001). Realized gains are higher in CRS than in DRS both for DA (MWU, N = 30, z = -4.500, p < 0.001) and for PO (MWU, N = 30, z = -2.344, p = 0.019).

In summary, we find that DA is able to realize a larger fraction of the maximal gains of trade than PO in both treatments, coming very close to the theoretical maximum. However, the difference is larger in the DRS treatment. To see this, we compare the difference in differences and find that the relative advantage of DA over PO is larger in DRS (24.0%) compared to CRS (14.6%; MWU, N = 30, z = 1.970, p = 0.049).

5.2 The Distribution of Gains from Trade

The previous subsection has examined efficiency at the level of an institution. In this subsection, we briefly compare the gains from trade separately for buyers and sellers, which illuminates the underlying dynamics. Figure 6 displays the per capita gains from actual trade over time in both treatments and both institutions, separately for buyers and for sellers (that is, the figure plots total gains of a market side divided by the number of actually-present traders from that side, averaged across active markets). While in DRS the gains from trade remain of a comparable magnitude for both market sides and for each given market institution, with sellers sightly above buyers, in CRS a large gap opens in DA, with buyers' gains steadily rising while those from sellers decline (the average



Figure 6: Average (over active markets) gains of buyers and sellers per period.

difference between per capita buyers' and sellers' gains is -2.0 ECUs in part 1 and 52.9 in part 3; WSR, N = 15, z = -3.408, p < 0.001). This gap also exists but is smaller in the case of PO (the corresponding average difference in part 3 for PO is 21.6, which is smaller than the one of DA; WSR, N = 15, z = -2.840, p = 0.005).

The interpretation is as follows. Over time, more traders move toward the double auction institution (Figure 3, top-right). In this institution, provided enough sellers are present, the price should converge to market clearing, which under constant returns to scale implies zero profits for the sellers (Figure 4, right). Convergence, however, takes time, and even more so in a setting where traders can switch away from a given institution. This means that profits remain positive for sellers even as more of them them flock to the double auction institution and the price in the latter drops. Hence, even as the price becomes relatively close to marginal cost and buyers' are able to reap larger and larger shares of the gains from trade, the institution remains comparatively attractive for sellers. Sellers are effectively facing a tradeoff between an institution (PO) that favors them but is gradually becoming empty, and a more efficient institution (DA) where their profits might eventually vanish but remain positive within a finite horizon.

6 Results: Individual Behavior

In this section we look at individual behavior regarding market selection. In particular, we look at market switching behavior, that is, behavior of traders who decide to leave the market they are currently at and select the other market for the next period. A *(market) switch* is a market choice where a subject selects a market for the next round that is different from the market he is currently at. A total of 19.8% of all market-choice decisions were switches, 21.1% in the case of sellers and 18.4% in the case of buyers (obviously excluding the first period).

The previous literature on experimental markets has often looked at simple behavioral rules of thumb (e.g., imitation or myopic best reply) within single markets. In particular, and following the theoretical model of Vega-Redondo (1997), a series of experimental Cournot oligopoly markets examined the role of imitation in symmetric settings where one market side is summarized by an aggregate demand function and all subjects play the role of producers. Those experiments generally found evidence in favor of imitative rules when information was centered on individual actions and profits, with possible shifts toward more forward-looking rules as myopic best reply when more information on the market structure was provided (e.g., Huck et al., 1999; Offerman et al., 2002; Apesteguía et al., 2007, 2010), although results depend on the number of players, the length of the experiment, and other factors (Bosch-Domènech and Vriend, 2003; Huck et al., 2004; Friedman et al., 2015; Oechssler et al., 2016). For instance, Huck et al. (2002) suggested that their data would be consistent with a mixture of best reply and imitation. Those experiments, however, all consider production behavior within a single market institution. The exception is Alós-Ferrer and Kirchsteiger (2015), who conducted an experiment on platform selection where, however, actual trade followed a reduced-form approach (that is, traders did not make actual trading decisions). Alós-Ferrer and Kirchsteiger (2015) found evidence in favor of switching behavior reflecting which institutions had the maximum observed payoffs for the trader's market side, which can be seen as a form of imitation.

Following this literature, we hypothesized that simple rules of thumb would be able to explain switching behavior in our data. We consider three behavioral rules that specify switches of a particular type depending on different (observable) market outcomes. A buyer (seller) following a *Best-Price* rule chooses the institution with lowest (highest) observed price in the current period, effectively chasing after the best observed price. To prevent trivial cases, the rule also prescribes to avoid institutions where no trade occurred, hence no price was realized. A buyer (seller) following a *Best-Average-Price* rule chooses the institution with the lowest (highest) average price in the current period, hence taking into account that a single price might not be representative of the institution (trivial cases are avoided as in Best-Price). A more forward-looking rule is to focus on the trader ratio as a predictor of future performance. In particular, in our setting, the equilibrium market price is a function of the trader ratio, and hence focusing on the

	Switches	Consistent decisions					
-		BestPrice		BestAv	BestAvgPrice		Ratio
Seller	21.1%	75.1%		69.4%		74.3%	
Buyer	18.4%	81.0%		77.7%		67.2%	
Consist	ent switches	Best	Price	BestAv	vgPrice	Best	Ratio
		DRS	CRS	DRS	CRS	DRS	CRS
Seller		61.9%	62.4%	63.6%	65.7%	77.1%	78.6%
Buyer		59.0%	59.6%	62.1%	66.5%	76.9%	73.3%

Table 5: Consistent decisions and consistent switches

Notes: BestAvgPrice, BestAvgPrice, BestRatio indicate that institution with best price, best average price (for own market side), or best buyer/seller ratio, respectively, was chosen. Imitation switches are switches in line with the respective rule.

latter could be seen as an attempt to predict the former, and in this sense this rule is closer to myopic best reply. A buyer following a *Best-Ratio* rule chooses the institution with lowest ratio of buyers to sellers in the current period (and follows the sellers if they are all at the same institution). Analogously, a seller following this rule chooses the institution with lowest ratio of sellers to buyers in the current period (and follows the buyers the buyers if they are all at the same institution).²¹

The upper part of Table 5 shows the percentage of decisions that are consistent with each rule's prescriptions, in addition to the percentage of switching decisions, separately for buyers and sellers. All three rules explain a large percentage of decisions, with the simplest one, BestPrice, explaining the most both for buyers (81.0%) and for sellers (75.1%), but with the most sophisticated, BestRatio, coming close for sellers (74.3%). Since decision inertia might explain a part of the non-switching decisions, the lower part of the table shows the fraction of *switches* that are consistent with following each rule, separately for trader type and treatment. BestRatio explains most switches (between 73.3% and 78.6%) for both trader types and in both treatments.

Given these results, we further analyze the predictive power of BestPrice and BestRatio in a series of random effects probit regressions (Tables 6 and 7). The dependent variable is switching behavior (a dummy variable taking the value 1 if there was a switch), but the regressors are continuous variables reflecting the price or ratio differences. The regression hence considers stochastic versions of the rules where a switch is more likely the larger the price difference (or ratio difference) is. The difference in the best price is coded as the difference between the currently chosen institution and the other institution for buyers and conversely for sellers. The difference in ratios is coded as the difference in buyer-seller ratios between the current institution and the other one for buyers, and the analogous difference in seller-buyer ratios for sellers. To avoid the natural (nonlinear)

²¹Note that the highest price and the highest average price is displayed explicitly in the market selection stage, whereas the ratio of buyers to sellers is not shown explicitly but can be computed fairly easy as both the number of buyers and sellers at each institution is shown on screen.

asymmetry of ratios (all ratios toward one side are condensed between 0 and 1, while ratios toward the other side are in principle unbounded), we consider the logarithm of the ratio. Last, to make the coefficients of the different rules comparable, we divide each of the three resulting variables by their respective empirical standard deviation.²² We remark that price differences and log of ratios are undefined if either institution is empty in a given period and hence the regressions must exclude those observations. However, in those cases all rules would make the same prediction, thus they play no role to compare the rules (the regressions, however, underestimate the predictive power of the rules for that reason).

The regressions show that both BestPrice and BestRatio are predictive of switches for both trader types if considered separately. However, when taken together, in the case of buyers only BestPrice remains significant. That is, buyer behavior appears to be usefully summarized by chasing after low prices. In contrast, for sellers both rules remain significantly predictive when considered jointly. That is, sellers seem to take into account both past high prices and favorable trader ratios, suggesting a stronger focus on forward-looking evaluations. Notably, this observation holds for both treatments, DRS and CRS.

The striking asymmetry in behavioral drivers between buyers and sellers is natural. In principle, in DA and under decreasing returns to scale, both trader types are treated symmetrically. However, PO is biased in favor of the sellers, and buyers are essentially passive in this latter institution. Thus, there is an institutional asymmetry which creates a tradeoff for sellers which is absent for the case of buyers. On the one hand, sellers would like to reap the high profits associated with larger prices typical of PO. On the other hand, the inefficiency of the latter institution attracts less buyers, thus leading to worse ratios for the sellers. This also holds empirically for constant returns to scale, even though in this case coordination on DA would make the sellers' profits disappear entirely. The latter observation is less surprising in view of previous sections, since the slow speed of convergence guarantees that profits in DA, although dwindling, remain positive for sellers.

It is also interesting to note that previous empirical work on experimental markets (within a fixed institution, and with respect to trading behavior) has often suggested behavioral heterogeneity, with some subjects relying more on imitative rules and others on myopic best reply (Huck et al., 1999, 2002), or behavior reflecting more or less sophisticated rules depending on available information (Offerman et al., 2002). In our markets, all traders have the same information, and they are randomly allocated to the role of buyers or sellers. Thus, the difference in behavior that we observe between buyers and sellers is causally induced by their different roles (and, specifically, the presence

 $^{^{22}}$ That is, we essentially standardize the difference variables, dividing the values by the standard deviation in the 30×24 dataset containing the price and trader-ratio differences in the experiment. We do not subtract the empirical mean because the value zero has a specific meaning (same prices or same ratios in both institutions).

Switch		DRS			CRS	
-	1	2	3	4	5	6
DiffBestPrice	0.268^{***}		0.266^{***}	0.358^{***}	k	0.365***
	(0.026)		(0.028)	(0.038)		(0.041)
DifflogRatio		0.197^{***}	6 0.006		0.148^{***}	-0.016
		(0.041)	(0.046)		(0.035)	(0.040)
Period	-0.032^{***}	-0.035^{***}	-0.032^{***}	-0.044^{**}	* -0.048***	-0.043^{***}
	(0.006)	(0.006)	(0.006)	(0.005)	(0.005)	(0.005)
Constant	-0.229^{***}	-0.396^{***}	-0.232^{**}	-0.293^{**}	* -0.324***	-0.288^{***}
	(0.089)	(0.095)	(0.091)	(0.080)	(0.081)	(0.081)
Observations	1350	1406	1350	1873	1929	1873

Table 6: Random effect probit regression on switches for buyers.

Notes: Standard errors in parentheses. * p < 0.1, ** p < 0.05, *** p < 0.01.

Table 7: Random effect probit regression on switches for sellers.

Switch		DRS				CRS	
—	1	2	3		4	5	6
DiffBestPrice	0.243^{***}		0.174^{***}		0.251***	*	0.130***
	(0.035)		(0.039)		(0.033)		(0.038)
DifflogRatio		0.278^{***}	* 0.190***			0.284^{***}	0.231^{***}
		(0.043)	(0.048)			(0.032)	(0.038)
Period	-0.044^{***}	-0.039^{***}	* -0.041***	_	-0.026***	* -0.020***	-0.022^{***}
	(0.007)	(0.007)	(0.007)		(0.005)	(0.005)	(0.005)
Constant	-0.375^{***}	-0.497^{***}	-0.448^{***}	-	-0.226***	* -0.353***	-0.308^{***}
	(0.089)	(0.089)	(0.091)		(0.068)	(0.068)	(0.069)
Observations	1349	1405	1349		1877	1933	1877

Notes: Standard errors in parentheses. * p < 0.1, ** p < 0.05, *** p < 0.01.

of an institution which treats them differently), and does not arise from cognitive or informational differences.

7 Conclusion

Most goods can be traded in different market institutions or platforms. The economics literature has intensively studied trading when the market institution is fixed, but market selection has received comparatively little attention. We study both, market selection itself, and how the presence of alternative institutions affects trade. We pit a doubleauction environment, where buyers and sellers can freely trade without an auctioneer, against a posted-offers institution, where buyers are constrained to accept or reject the offers made by sellers. While the former has been empirically shown to induce quick convergence to market clearing in isolation, the latter embodies a bias in favor of the sellers, and thus efficiency arguments suggest that traders should learn to coordinate on the double auction.

We observe coordination toward the double-auction institution, which is faster when sellers face decreasing returns to scale compared to when they are endowed with constant per-unit production costs. That is, under decreasing returns to scale, traders learn to coordinate on the efficient double auction, and the posted-offers institution becomes inactive in finitely many rounds. In contrast, under constant returns to scale, although traders tend to concentrate on the double auction, both institutions remain active until the last round. The reason is that, in the latter case, sellers try to stave off (but ultimately are not able to prevent) coordination in the double auction and the resulting convergence to market clearing prices (which would leave them with no profits).

Within institutions, and even though traders might "jump boat" at any point, trading behavior approaches equilibrium predictions in terms of prices and gains of trade. Switching behavior across institutions, however, can be explained by simple rules of thumb. Those are different across trader types: buyers seem to simply chase after low observed prices, while sellers consider both high observed prices and trader ratios. The latter might be seen as a myopic predictor of future market prices and hence reveal a more forward-looking mindset. Since traders were randomly allocated to the roles of buyers or sellers, the differences in behavior were causally induced.

Our work empirically demonstrates that efficiency alone might not suffice for a market institution to drive competing ones out of the field. Under decreasing returns to scale, fast coordination occurs because the double auction Pareto dominates the alternative, in the sense that both trader types can reap larger gains from trade. Under constant returns to scale, gains from trade are monopolized by buyers, and sellers' earnings approach zero as prices approach the market clearing level. Thus, the distribution of gains from trade is unequal and sellers do not benefit from coordination on the efficient institution. Although, in our data, this translated into a slow-down of convergence, in general one might speculate that once profits become small enough, sellers might, in some cases and for some parameter constellations, successfully manage to turn the tide and preserve biased, inefficient institutions in the market. This might contribute to explain why multiple, possibly-inefficient institutions typically coexist for single, given goods in actual markets.

References

- Alós-Ferrer, C. and Kirchsteiger, G. (2015). Learning and Market Clearing: Theory and Experiments. *Economic Theory*, 60(2):203–241.
- Alós-Ferrer, C. and Kirchsteiger, G. (2017). Market Selection by Boundedly-Rational Traders under Constant Returns to Scale. *Economics Letters*, 153:51–53.
- Alós-Ferrer, C., Kirchsteiger, G., and Walzl, M. (2010). On the Evolution of Market Institutions: The Platform Design Paradox. *Economic Journal*, 120(543):215–243.

- Apesteguía, J., Huck, S., and Oechssler, J. (2007). Imitation—Theory and Experimental Evidence. Journal of Economic Theory, 136:217–235.
- Apesteguía, J., Huck, S., Oechssler, J., and Weidenholzer, S. (2010). Imitation and The Evolution of Walrasian Behavior: Theoretically Fragile But Behaviorally Robust. *Journal of Economic Theory*, 145:1603–1617.
- Ariely, D., Ockenfels, A., and Roth, A. (2005). An Experimental Analysis of Ending Rules in Internet Auctions. *RAND Journal of Economics*, 36:890–907.
- Ausubel, L. M., Cramton, P., Pycia, M., Rostek, M., and Weretka, M. (2014). Demand Reduction and Inefficiency in Multi-Unit Auctions. *Review of Economic Studies*, 81(4):1366–1400.
- Bosch-Domènech, A. and Vriend, N. J. (2003). Imitation of Successful Behaviour in Cournot Markets. *Economic Journal*, 113(487):495–524.
- Campbell, J., LaMaster, S., Smith, V. L., and Van Boening, M. (1991). Off-Floor Trading, Disintegration, and the Bid–Ask Spread in Experimental Markets. *Journal* of Business, 64:495–522.
- Chamberlin, E. H. (1948). An Experimental Imperfect Market. Journal of Political Economy, 56(2):95–108.
- Crockett, S., Friedman, D., and Oprea, R. (2020). Naturally Occurring Preferences and General Equilibrium: A Laboratory Study. Discussion Paper.
- Crockett, S., Oprea, R., and Plott, C. R. (2011). Extreme Walrasian Dynamics: The Gale Example in the Lab. *American Economic Review*, 101(7):3196–3220.
- Fischbacher, U. (2007). z-Tree: Zurich Toolbox for Ready-Made Economic Experiments. Experimental Economics, 10(2):171–178.
- Friedman, D., Huck, S., Oprea, R., and Weidenholzer, S. (2015). From Imitation to Collusion: Long-Run Learning in a Low-Information Environment. *Journal of Economic Theory*, 155:185–205.
- Genesove, D. and Hansen, J. (2016). The Role of Auctions and Negotiations in Housing. CEPR Discussion Paper 11392.
- Gillen, B. J., Hirota, M., Hsu, M., Plott, C. R., and Rogers, B. W. (2020). Divergence and Convergence in Scarf Cycle Environments: Experiments and Predictability in the Dynamics of General Equilibrium Systems. *Economic Theory*, forthcoming.
- Greiner, B. (2015). Subject Pool Recruitment Procedures: Organizing Experiments with ORSEE. Journal of the Economic Science Association, 1:114–125.
- Hayek, F. (1967). *Studies in Philosophy, Politics and Economics.* Routledge & Paul Kegan, London.
- Huck, S., Normann, H.-T., and Oechssler, J. (1999). Learning in Cournot Oligopoly -An Experiment. *Economic Journal*, 109:C80–C95.
- Huck, S., Normann, H.-T., and Oechssler, J. (2002). Stability of the Cournot Process — Experimental Evidence. *International Journal of Game Theory*, 31:123–136.

- Huck, S., Normann, H.-T., and Oechssler, J. (2004). Two Are Few and Four Are Many: Number Effects in Experimental Oligopolies. *Journal of Economic Behavior and Or*ganization, 53(4):435–446.
- Klein, B., C. R. and Alchian, A. (1978). Vertical Integration, Appropriable Rents, and the Competitive Contracting Process. Journal of Law and Economics, 21(2):297–326.
- Kugler, T., Neeman, Z., and Vulkan, N. (2006). Markets versus Negotiations: An Experimental Investigation. Games and Economic Behavior, 56(3):121–134.
- Lucking-Reiley, D. (2000). Auctions on the Internet: What's Being Auctioned, and How? Journal of Industrial Economics, 48(3):227–252.
- Madhavan, A. (1992). Trading Mechanisms in Securities Markets. *Journal of Finance*, 47(2):607–641.
- Oechssler, J., Roomets, A., and Roth, S. (2016). From Imitation to Collusion: A Replication. Journal of the Economic Science Association, 2:13–21.
- Offerman, T., Potters, J., and Sonnemans, J. (2002). Imitation and Belief Learning in an Oligopoly Experiment. *Review of Economic Studies*, 69(4):973–997.
- Plott, C. R. (1982). Industrial Organization Theory and Experimental Economics. Journal of Economic Literature, 20:1485–1587.
- Plott, C. R. (2000). Market Stability: Backward-Bending Supply in a Laboratory Experimental Market. *Economic Inquiry*, 38:1–18.
- Plott, C. R. and Gray, P. (1990). The Multiple Unit Double Auction. Journal of Economic Behavior and Organization, 13:245–258.
- Plott, C. R. and Smith, V. L. (1978a). An Experimental Examination of Two Exchange Institutions. *Review of Economic Studies*, 45:133–153.
- Plott, C. R. and Smith, V. L., editors (1978b). Handbook of Experimental Economics Results, Volume 1. North Holland, Amsterdam.
- Quan, D. C. (1994). Real Estate Auctions: A Survey of Theory and Practice. Journal of Real Estate Finance and Economics, 9:23–49.
- Reynolds, S. and Wooders, J. (2009). Auctions with a Buy Price. *Economic Theory*, 38:9–39.
- Rustichini, A., Satterthwaite, M. A., and Williams, S. R. (1994). Convergence to Efficiency in a Simple Market with Incomplete Information. *Econometrica*, 62(5):1041–1063.
- Smith, V. L. (1962). An Experimental Study of Competitive Market Behavior. Journal of Political Economy, 70(2):111–137.
- Van Huyck, J., Battalio, R. C., and Rankin, F. (1997). On the Origin of Convention: Evidence from Coordination Games. *Economic Journal*, 107(442):576–597.
- Vega-Redondo, F. (1997). The Evolution of Walrasian Behavior. *Econometrica*, 65(2):375–384.

- Williams, A., Smith, V. L., Ledyard, J., and Gjerstad, S. (2000). Concurrent Trading in Two Experimental Markets with Demand Interdependence. *Economic Theory*, 16:511–528.
- Williamson, O. (1975). Markets and Hierarchies: Analysis and Antitrust Implications. Free Press, New York.
- Young, H. P. (1993). The Evolution of Conventions. Econometrica, 61:57-84.
- Zumpano, L. V., Elder, H. W., and Baryla, E. A. (1996). Buying a House and the Decision to Use a Real Estate Broker. *Journal of Real Estate Finance and Economics*, 13:169–181.

Appendix A Additional Analyses

	Number of	buyers at DA	Number of	sellers at DA
Period	0.0735^{***}	0.0666^{***}	0.0971^{***}	0.0902^{***}
	(0.0241)	(0.0191)	(0.0346)	(0.0274)
L.BSRatioDA		-0.3049^{*}		0.1952
		(0.1395)		(0.1617)
Constant	0.8586^{***}	1.2445^{***}	0.7574^{***}	0.5732^{*}
	(0.1566)	(0.2596)	(0.1859)	(0.2754)
Observations	375	359	375	359
Markets	15	15	15	15

Table A.1: Fractional Logit regressions for DRS.

Notes: Robust standard errors in parentheses. * p < 0.1, ** p < 0.05, *** p < 0.01.

Table A.2. Tractional Logit regressions for City.								
	Number of	buyers at DA	Number of	Number of sellers at DA				
Period	0.0678^{***}	0.0646^{***}	0.0400^{***}	0.0355^{***}				
	(0.0127)	(0.0114)	(0.0106)	(0.0100)				
L.BSRatioDA		-0.2898^{**}		0.2374^{**}				
		(0.0889)		(0.1147)				
Constant	0.5332^{***}	0.8752^{***}	0.5386^{***}	0.3475				
	(0.1453)	(0.2336)	(0.1194)	(0.2027)				
Observations	375	360	375	360				
Markets	15	15	15	15				

Table A.2: Fractional Logit regressions for CRS.

Notes: Robust standard errors in parentheses. * p < 0.1, ** p < 0.05, *** p < 0.01.

Table A.3: Fractional Logit regressions for comparison DRS vs CRS.

	Number of	buyers at DA	Number of	Number of sellers at DA		
	Rounds 2-9	Rounds $18-25$	Rounds 2-9	Rounds $18-25$		
Period	0.0968^{***}	0.0471	0.1061^{***}	0.0251		
	(0.0309)	(0.0302)	(0.0312)	(0.0249)		
CRS	-0.4033	-0.6052	-0.3294	-1.0557^{**}		
	(0.2440)	(0.5039)	(0.2447)	(0.5233)		
Constant	0.7169^{***}	1.4100^{***}	0.4806^{***}	1.7312^{***}		
	(0.1851)	(0.5460)	(0.1722)	(0.6562)		
Observations	240	240	240	240		
Markets	30	30	30	30		

Notes: Robust standard errors in parentheses. * p < 0.1, ** p < 0.05, *** p < 0.05.



Figure A.1: Number of buyers/sellers at DA per period, separately for each individual market observation in DRS.



Figure A.2: Number of buyers/sellers at DA per period in DRS, separately for each individual market observation in CRS.

Appendix B	Equilibrium	outcomes
------------	-------------	----------

$n \mid m$	1	2	3	4	5	6	7	8
1	[50, 55]	40	35	35	35	[15, 35]	15	15
2	60	[50, 55]	40	40	[35, 40]	35	35	35
3	65	60	[50, 55]	[40, 50]	40	40	40	35
4	65	60	[55, 60]	[50, 55]	50	40	40	40
5	65	[60, 65]	60	55	[50, 55]	50	40	40
6	[65, 80]	65	60	60	55	[50, 55]	50	[40, 50]
7	80	65	60	60	60	55	[50, 55]	50
8	80	65	65	60	60	$[55,\!60]$	55	[50, 55]

Table B.1: Equilibrium price matrix for n buyers and m sellers for DRS.

$n \mid m$	1	2	3	4	5	6	7	8
1	40	40	40	40	40	40	40	40
2	[55, 60]	40	40	40	40	40	40	40
3	[60, 65]	[40, 55]	40	40	40	40	40	40
4	65	[55, 60]	40	40	40	40	40	40
5	65	60	55	40	40	40	40	40
6	[65, 80]	[60, 65]	[55, 60]	[40, 55]	40	40	40	40
7	80	65	60	55	40	40	40	40
8	80	65	60	$[55,\!60]$	55	40	40	40

Table B.2: Equilibrium price matrix for n buyers and m sellers for CRS.

n / m	1	2	3	4	5	6	7	8
1	120	160	185	205	225	245	245	245
2	150	240	290	320	350	370	450	530
3	170	275	360	420	450	480	510	535
4	185	300	400	480	540	580	610	640
5	200	325	425	520	600	660	710	740
6	215	340	450	550	640	720	780	840
7	215	355	475	575	675	760	840	900
8	215	370	495	600	700	800	880	960

Table B.3: Maximum possible gains of trade matrix for n buyers and m sellers for DRS.

n / m	1	2	3	4	5	6	7	8
1	100	100	100	100	100	100	100	100
2	170	200	200	200	200	200	200	200
3	195	300	300	300	300	300	300	300
4	210	340	400	400	400	400	400	400
5	225	365	470	500	500	500	500	500
6	240	390	510	600	600	600	600	600
7	240	405	535	640	700	700	700	700
8	240	420	560	680	770	800	800	800

Table B.4: Maximum possible gains of trade matrix for n buyers and m sellers for CRS.

Appendix C Screenshots of the trading interface

										Marke	et A						Mar	ket B		
			Mark	et A	Mark	et B														
Number buyers	3		5		3		Qua	ntity						Qı	antity 3 ₁					
Number sellers	•		2		6															
Average price per unit tradec	I		5'		5	2	2		L						2					
Average quantity tr per buyer	aded		1.	6	2.	D	1-	1				÷.			1	÷.		T.		
Average quantity tr by seller	aded		4.)	1.	D	0	45			55		65		0	45	50		60	
					1					Pric	e						F	rice		
Your re	esults of	this tra	ding day:							On whi	ch mar	ket wou	ld you like	e to tra	de durir	ig the n	ext tradin	g day?		
You were trading on market	Α.																			
Your units	1	2	3 4		5 6							Go to n	narket A			Go	to marke	t B		
Price of your traded units	60													_						

Figure C.1: Interface for market selection stage.

Figure C.2: Interface for double auction institution.

Trading Day 1: Y	ou are trading on n	narket A. There are 5 buyers and 4 s	seller	rs active on market A.		Remaining time [sec]: 30
Your offer to buy (46) Offers to buy by other buyers (47) 47) 47) 47) 47) 47) 47) 47)	Remaining u can still buy: Sub Pri Quant	rou are a buyer. nits, that you 5 mit an offer to buy ce 46 ity 2 Submit offer Withdraw offer		0(ffers for sale 60) 0(60) 55 65 65 65 65 67	Accept ar To pri sh	n offer buy a unit from a seller at the offered ce, please click on the corresponding box owing that price.
Resale values Resale value of 1st unit Resale value of 2nd unit Resale value of 3rd unit Resale value of 4th unit Resale value of 5th unit Resale value of 6th unit	80 65 60 55 40 35	Price of your bought units 46		If you decide to leave the trading stag further trade. The trading stage ends if no further tra	e early, ti de is pos	hen you will not be able to conduct any asible or when the time is over.

Trading Day 1	. Tou are trading on			
Your offer for sale		You are a seller		
50 §Q 60	Remaining still sell:	units that you can 6		
	Sul	bmit an offer for sale		
	Quar	ntity 3		
		Submit offer Withdraw offer		
		Price of your sold units		
Production costs			If you decide to leave the offer stage early	, then you will not be able to change your offer.
Cost of the 1st unit	15		The offer stage ends if all sellers have left	the offer stage or when the time is over.
Cost of the 2nd unit	35			
Cost of the 3rd unit	40			
Cost of the 4th unit	50			
Cost of the 5th unit	60			
	65			Leave offer store early
Cost of the 6th unit	: You are trading on	market B. There are 3 buyers and 4 s	sellers active on market B.	Remaining time [sec]: 20
Cost of the 6th unit	: You are trading on	market B. There are 3 buyers and 4 s	sellers active on market B.	Remaining time (sec): 20
Cost of the 6th unit Trading Day 1	: You are trading on Remaining can still buy	market B. There are 3 buyers and 4 s You are a buyer. units that you 5	sellers active on market B. Offers for sale 60 90 60 96 55 65 65 67	Remaining time [sec]: 20 apt an offer 20 apt an offer 20 To buy a unit from a seller at the offered price, please click on the corresponding box showing that price. 20
Cost of the 6th unit Trading Day 1 Resale values	CS : You are trading on Remaining can still buy	market B. There are 3 buyers and 4 s You are a buyer. units that you 5 ?	sellers active on market B. Offers for sale 60 0 60 55 65 65 67 If you decide to leave the trading stage ea	Remaining time [sec]: 20 ept an offer 20 apt an offer 20 To buy a unit from a seller at the offered price, please click on the corresponding box showing that price. 20 rhy, then you will not be able to conduct any 20
Cost of the 6th unit Trading Day 1 Resale values Resale values Resale values	COS You are trading on Remaining can still buy 80	market B. There are 3 buyers and 4 s You are a buyer. units that you 5 	Sellers active on market B. Offers for sale 00 00 80 55 65 65 65 65 65 87 If you decide to leave the trading stage ear further trade.	Remaining time [sec]: 20 apt an offer 20 apt an offer 20 To buy a unit from a seller at the offered price, please click on the corresponding box showing that price. 20
Cost of the 6th unit Trading Day 1 Trading Day 1 Resale values Resale values Resale value of the 1st unit Resale value of the 2nd unit	: You are trading on Remaining can still buy 80 65	market B. There are 3 buyers and 4 s You are a buyer. units that you 5 ?	ellers active on market B.	Remaining time [sec]: 20 apt an offer 20 apt an offer 20 Now it is your turn. You can buy units that are still available. 20 To buy a unit from a selier at the offered price, please click on the corresponding box showing that price. 20 rrly, then you will not be able to conduct any s possible or when all traders had their turn. 20
Cost of the 6th unit Trading Day 1 Trading Day 1 Resale values Resale values Resale value of the 1st unit Resale value of the 2nd unit Resale value of the 3nd unit	You are trading on Control of the second s	market B. There are 3 buyers and 4 s You are a buyer. units that you 5	sellers active on market B.	Remaining time [sec]: 20 apt an offer 20 apt an offer 20 Now it is your turn. You can buy units that are still available. 20 To buy a unit from a seller at the offered price, please click on the corresponding box showing that price. 20 rly, then you will not be able to conduct any s possible or when all traders had their turn. 20
Cost of the 6th unit Trading Day 1 Trading Day 1 Resale values Resale values Resale value of the 1st unit Resale value value of the 1st unit Resale value value of the 1st unit Resale value	Co Co	market B. There are 3 buyers and 4 s You are a buyer. units that you 5 Price of your bought units 55	sellers active on market B.	Remaining time [sec]: 20 apt an offer 20 apt an offer 20 Now it is your turn. You can buy units that are still available. 20 To buy a unit from a seller at the offered price, please click on the corresponding box showing that price. 20 rdy, then you will not be able to conduct any s possible or when all traders had their turn. 20
Cost of the 6th unit Trading Day 1 Trading Day 1 Resale values Resale value of the 1st unit Resale value of the 1st unit Resale value of the 3rd unit Resale value of the 4th unit Resale value of the 5th unit Resale value value of the 5th unit Resale value of the 5th unit Resale value value value of the 5th unit Resale value value of the 5th unit Resale value value of the 5th unit Resale value value value of the 5th unit Resale value	Cos C	market B. There are 3 buyers and 4 s You are a buyer. units that you 5 Price of your bought units 55	sellers active on market B.	Remaining time [sec]: 20 apt an offer 20 apt an offer 20 To buy a unit from a seller at the offered price, please click on the corresponding box showing that price. 20 rty, then you will not be able to conduct any s possible or when all traders had their turn. 20

Figure C.3: Interface for posted offer institution. Top: Sellers. Bottom: Buyers.

Appendix D Translated Instructions

The original instructions were in German. Text in brackets [...] was not displayed to subjects.

General

Welcome. This is an experiment on the economics of market decision making. The instructions are simple and if you follow them carefully and make good decisions you might earn a considerable amount of money. Your earnings depend on your decisions and the decisions made by other participants in the experiment.

In this experiment you will earn experimental currency units (ECU). At the end of the experiment all ECU you have earned throughout the experiment will be added up and the sum total will be converted to EURO according to the following exchange rate:

100 ECU = 1.50 EUR.

At the end of the experiment, your earnings will be paid to you in cash. Independently of your decisions, you will receive an additional 4 EUR for your participation in the experiment. The experiment will take about two hours.

Course of the experiment

The experiment consists of two parts, a market experiment and an second part with multiple decision situations. Your earnings in each part are independent of each other. Detailed instructions for each part will be provided at the beginning of each part.

Instructions for market experiment

The market experiment consists of 2 trial rounds and 25 trading days (rounds), on which you can trade units of a good on two different markets. For this part of the experiment you are randomly assigned to a group of 16 participants (including yourself). Half of this group (8) will act as sellers and the other half (8) will act as buyers. Your role, the role of the other participants and the composition of your group will remain fixed for the entire duration of the market experiment. In particular, each group member either acts always as a seller or always as a buyer. All sellers sell the same good and all buyers can buy only this good.

Each trading day consists of two stages: First, there is a market selection stage and then a trading stage.

In the market selection stage, you can decide on which of the two markets you want to trade. During a trading day you can only trade at a single market, that is, in the trading stage you can only trade on the market you selected. You can switch the market again in the market selection stage of the next trading day. In the trading stage, you can trade at your selected market. Each trader (buyer or seller) can trade up to 6 units of the good in every trading stage. That is, on each trading day a seller can sell up to 6 units and a buyer can buy up to 6 units.

There are two markets that differ in the market procedure. In what follows, we will simply refer to those markets as market A and market B.

First, there are two trial rounds so that you can familiarize yourself with the market interface and the details of the trading procedures. In the trial rounds, there is no market selection stage but instead 4 buyers and 4 sellers are automatically assigned to market A in one trial round and to market B in the other trial round.

The trial rounds have no consequences for your earnings in the market experiment. Only the following 25 trading days determine your payoff for the market experiment.

Course of action for the market selection stage

In the market selection stage, you can choose on which of the two markets, market A or market B, you want to trade during the next trading day. Additionally you are presented with an overview of the results of the previous trading day for both markets. The market selection stage ends automatically after 30 seconds or when all 16 traders have selected a market. The remaining time is shown in the top right corner of the screen (see Figure 1 below). To select market A, simply click the button labeled "Go to market A." To select market B, simply click the button labeled "Go to market B."

If you fail to select a market within 30 seconds, then you can not trade on any market during the next trading day, hence you will earn no profits in that round.

In the market selection stage, the results of the previous trading day are displayed as follows (see Figure 1 below):

- In the upper left part of the screen, you can see the number of buyers, the number of sellers, the average price per unit traded, the average traded quantity per buyer and the average traded quantity per seller, separately for each market.
- The upper right part of the screen shows a price-quantity histogram, which shows the prices of all traded units and their frequency, separately for each market.
- In the lower left part of the screen you can see an overview of your own results of the previous trading day, that is, the prices of all your traded units as well as your profit for that trading day.



Figure 1. Example of market selection stage.

Specific instructions for buyers

[We present the instructions for buyers. The instructions for sellers were framed analogously. Instructions only differ in the description of market B, hence we provide both versions in that case.]

For the entire duration of the market experiment you are acting as a **BUYER**.

As a buyer, you can buy up to 6 units on each trading day. For each unit that you buy on a trading day you will receive a certain amount of ECU. You can see this resale value of a unit in the lower left part of the trading screen in the table "Resale values." We now explain how to read this table.

Resale values							
Resale value of 1st unit	190						
Resale value of 2nd unit	170						
Resale value of 3rd unit	150						
Resale value of 4th unit	140						
Resale value of 5th unit	130						
Resale value of 6th unit	110						

For the first unit that you sell on a trading day you receive the amount listed in the first row of the table labeled "Resale value of 1st unit." For the second unit that you sell on a trading day, you receive the amount listed in the second row labeled "Resale value of the 2nd unit," and so on. The values in this table only serve as an illustration and the actual values in the experiment will be different from those. The values in this table may change from trading day to trading day.

Your profit for each traded unit is the difference between the resale value of that unit and the price at which you have bought that unit.

Profit of a traded unit = Resale value of that unit - Price of that unit

Example: Suppose the resale values of the first and second unit are 190 and 170, respectively, as illustrated in the table above. For simplicity, in this example we only consider two units. If you buy the first unit at a price of 150 and the second unit at a price of 140, then your profit is as follows:

- Profit of the first unit = 190 150 = 40
- Profit of the second unit = 170 140 = 30

On the following pages, we will detail the trading procedure on market A and market B using exemplary figures. Of course, the exact numbers are only examples and the actual numbers you will encounter latter on in the experiment will be different from those. However, the structure and layout of the screen will be the same in the actual experiment.

Trading procedure on market A [buyers]

A trading phase on market A can last up to 90 seconds. The remaining time is shown in the top right corner of the screen (see Example A below). The trading phase ends automatically when the time is up or when no further trade is possible (e.g. because all buyers have left the trading phase).

At market A, buyers and sellers can make offers and accept offers of other traders during the entire trading phase. At market A, as a buyer you have two ways to trade: First, you can submit an offer to buy. Sellers can then accept your offer and trade units with you at the price you offered to buy. Second, you can accept open offer for sale of a seller and in that way buy units at the offered price.

Each trader can only have one active offer, that is, you cannot have multiple active offers at the same time. You can submit a new offer at any time. In that case, your current offer will be replaced by the new offer. Additionally, you can withdraw a standing offer by pressing the "Withdraw offer" button.

Trading Day 1: Y	ou are trading on m	arket A. There are 5 buyers and 4 s	seller	rs active on market A.		Remaining time [sec]: 30
Your offer to buy 46 Offers to buy by other buyers 47 47 47 47 47 47 47 47 66 60	Yu Remaining un can still Duy: Subr Price Quanti	its, that you 5 mit an offer to buy the 46 ty 2 Submit offer /ithdraw offer		Ac Offers for sale 60 90 60 96 55 65 65 67	To prish	n offer buy a unit from a seller at the offered ce, please click on the corresponding box owing that price.
Resale values Resale value of 1st unit Resale value of 2nd unit Resale value of 3rd unit Resale value of 3th unit Resale value of 5th unit Resale value of 6th unit	80 65 60 55 40 35	Price of your bought units 46		If you decide to leave the trading stage of further trade. Turther trade. The trading stage ends if no further trade	early, th	hen you will not be able to conduct any ssible or when the time is over. Leave trading stage early

Example A. Trading interface for market A.

How can I accept an active offer for sale?

Active offers for sale are shown in the upper right part of the trading interface. Each row shows the active offers for sale of one particular seller, where the order of sellers is random. The number of boxes shows the number of units that are offered, that is, each box represents an offer for sale for a single unit. You can buyer an offered unit from a seller by clicking on the corresponding box. A unit that is still available is shown as a gray box. Already traded units are crossed out to indicate that they are not available anymore for trade.

You can only accept an offer for sale if you can buy that unit without making losses. Here, previous profits during the same trading day are also taken into account, that is, the sum of the resale value of the next unit and your previous profits needs to be larger than the price of the unit you want to buy.

How can I submit an offer to buy?

You and all other buyers can submit an offer to buy. An overview of all active offers to buy is shown in the upper left part of the trading interface. Your own active offer is shown in the first row. The active offers of other buyers are shown directly below in random order. An offer consists of a price and a quantity (number of units) that you want to buy at that price. In the field "Submit an offer to buy" you can enter a price and a quantity and submit your offer by pressing the "Submit offer" button. You can only enter prices that are positive integer values, that is a price of 152 is possible but not a price of 152.5. On each trading day, you can not trade more than 6 units, that is, if you have already bought 2 units, then you can only submit offers with a quantity of at most 4.

Which offers to buy are valid?

You can only submit an offer, if you can buy all offered units at the offered price without making losses. Here, previous profits are taken into account, that is, an offer is valid if the sum of the resale values for all offered units and your previous profits is larger than the price of buying all offered units at the offered price (quantity times price).

What happens to my current active offer if I accept an offer for sale by a seller?

If you have submitted an offer to buy and afterwards you accept an offer by a seller by clicking on a unit (gray box), then it will be automatically checked whether your current offer is still valid. If it is not valid anymore, but would be valid with a lower quantity, then the quantity of your offer is automatically reduced accordingly. If there is no offer with that price that is valid, then you offer will be withdrawn automatically.

Trading procedure on market B [buyers]

At market B, only sellers can make offers and only buyers can accept active offers for sale made by selles. At market B, as a buyer you can only accept active offers.

First, all sellers simultaneously submit an offer for sale within 20 seconds. At market B, you and all other buyers are passive during that time. Afterwards, you and all other buyers are informed about all submitted offers for sale.

Next, you and the other buyers one after the other have the opportunity to buy units that were offered by the sellers. The order in which buyers take turns is determined randomly. The first buyer, can choose up to 6 units that he wants to buy from all submitted offers. Once the time is up or once the first buyers has traded all 6 units, it is the second buyer's turn, who also can buy up to 6 units. And so on. The trading phase ends if all buyers had their turn, or if all offered units were bought.

When it is your turn, you have 20 seconds to buy still available units. The remaining time is shown in the top right corner of the screen (see Example B below). If you do not buy any unit before the time is up, then you will make no profits on that trading day.

How can I accept an active offer for sale?

You can accept an offer for sale exactly as in market A (see above).

Trading Day 1: Y	'ou are trading on r	market B. There are 3 buyers and 4 s	ellers active on market B.	Remaining time [sec]: 20
	Remaining u can still buy	You are a buyer. inits that you 5	Offers for sale	Accept an offer Now it is your turn. You can buy units that are still available. To buy a unit from a seller at the offered price, please click on the corresponding box showing that price.
Resale values Resale value of the 1st unit Resale value of the 2nd unit Resale value of the 3rd unit Resale value of the 4th unit Resale value of the 5th unit Resale value of the 6th unit	80 65 60 55 40 35	Price of your bought units 55	If you decide to leave the trading s further trade. The trading stage ends if no further	tage early, then you will not be able to conduct any trade is possible or when all traders had their turn. Leave market stage early

Example B. Trading interface for market B.

Trading procedure on market B [sellers]

At market B, only sellers can make offers and only buyers can accept active offers for sale made by sellers. At market B, as a seller you can only submit offers for sale.

First, you and all other sellers simultaneously submit an offer for sale within 20 seconds. At market B, buyers are passive during that time. The remaining time is shown in the top right corner of the screen (see Example B below). During that time you can change your offer at any time and submit a new offer with a different price and/or quantity. Your current offer is replaced by the new offer. Additionally, within the time limit you can withdraw your offer by pressing the "Withdraw offer" button.

When the time is up, your current offer is submitted automatically. Once it is submitted you will not be able to change your offer during that trading day. In case you have not submitted an offer within the time limit or you have withdrawn your offer and not submitted a new offer, then you will make no profits on that trading day.

Once the time is up, buyers are informed about all submitted offers for sale. Then one after the other buyers have the opportunity to buy units that were offered for sale. The order in which buyers take turns is determined randomly. As a seller you can monitor trade on market B but remain passive. The trading phase ends if all buyers had their turn, or if all offered units were bought.

When it is your turn, you have 20 seconds to buy still available units. The remaining time is shown in the top right corner of the screen (see Example B below). If you do not buy any unit before the time is up, then you will make no profits on that trading day.

How can I submit an offer for sale?

You can submit an offer exactly as in market A (see above).

Trading Day 1: Yo	u are trading on ma	rket B. There are 3 buyers and 4	sellers active on market B.	Remaining time [sec]: 20
Your offer for sale	Yo	u are a seller.		
60 60 60	Remaining unit still sell:	s that you can 6		
	Submit	an offer for sale		
	Price	60		
	Quantity	3		
		ubmit offer		
	Wi	thdraw offer		
		Price of your sold units		
Production costs			If you decide to leave the offer stage early, the	en you will not be able to change your offer.
Cost of the 1st unit	15		The offer stage ends if all sellers have left the	offer stage or when the time is over.
Cost of the 2nd unit	35			
Cost of the 3rd unit	40			
Cost of the 4th unit	50			
Cost of the 5th unit	60			
Cost of the 6th unit	65			Leave offer stage early
				Louis one suga curry

Example B. Trading interface for market B.

Payoffs

We now explain how your earnings in the market experiment are determined. If you bought no unit on a trading day, then your profit for that trading day is zero. Your profit for each unit bought is the difference between the resale value of that unit and the price at which you have bought that unit.

Profit of a traded unit = Resale value of that unit - Price of that unit

Your profit on a trading day is the sum of all your profits from all units you have bought. At the end of the last trading day, your profits from all trading days are added up and converted to EURO (100 ECU = 1.50 EUR). This amount is your payoff for the market experiment.

Comprehension Questions

Please answer the following comprehension questions to check your understanding of the instructions.

1. Suppose you as a buyer have offered to buy 4 units at a price of 90 and a seller has accepted your offer for two units, that is, you have traded two units at that price. Suppose your resale value for the first and second unit is 130 and 110, respectively (you have not traded any other units so far).

What is your profit from trading those two units?

2. As a buyer in market A you can

- \bigcirc only accept offers.
- \bigcirc only submit offers.
- $\bigcirc\,$ accept and submit offers.
- 3. As a buyer in market B you can
 - \bigcirc only accept offers.
 - $\bigcirc\,$ only submit offers.
 - $\bigcirc\,$ accept and submit offers.
- 4. Suppose in the trading phase you have not selected one of the two markets within the time limit of 30 seconds. In that case, in the next round you will
 - \bigcirc trade at market A.
 - $\bigcirc\,$ trade at market B.
 - \bigcirc not be able to trade.